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Abstract 

The present thesis focuses on the characterization of dynamics of dispersed gas-liquid 

flow and liquid-phase mixing generated by multiple meandering bubble plumes in a scaled-

down cold flow model of Basic Oxygen Furnace (BOF) through measurements and CFD 

simulations. The role of complex interactions among meandering bubble plumes in the primary 

(in r-direction) and secondary (in θ-direction) flow, that governs the local mixing inside the 

vessel BOF vessel, was quantified. Cold flow experiments were performed to measure the 

instantaneous and time-averaged gas volume fraction, and local mixing time at different gas 

mass flow rates using in-house developed voidage and conductivity probes, respectively. 3D 

transient Euler-Lagrange (EL) simulations of dispersed gas-liquid flow were performed under 

cold flow conditions. The predicted instantaneous and time-averaged gas volume fraction and 

local as well as spatially-averaged mixing time were found in a satisfactory agreement with the 

measurements. The effect of oscillatory behavior of multiple bubble plumes on the dynamics 

of gas-liquid flow and mixing was studied. In the case of uniform flow scheme (conventional 

flow scheme used in industry), the liquid-phase flow field was found to be asymmetric and it 

led to asymmetric mixing inside the vessel. In this case, the local mixing time varied 

considerably from one location to the other location in the vessel. 

3D/transient Euler-Lagrange (EL) without/with Volume-of-Fluid (VOF) simulations of 

dispersed gas-liquid flow were performed to investigate the effect of the free gas-liquid 

interface on the dynamics of gas-liquid flow and mixing at different H/D ratios. For lower H/D 

ratios, EL simulations performed with no-slip and free-slip boundary conditions led to 

oscillatory plume behaviour and higher liquid velocity regions which in turn led to lesser 

mixing time. In contrast, EL+VOF simulations led to reduced meandering motion of bubble 
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plumes and lower liquid velocities resulting in higher mixing times. Interestingly, the mixing 

time predicted using the EL+VOF approach was found to be in a good agreement with the 

measurements 

The effect of PCD (pitch-to circle-diameter ratio), which provide the flow variation in 

the radial direction, on dynamics of gas-liquid flow and mixing was studied. It was found that 

the flow was symmetric when eight bottom tuyeres were placed at PCD of 0.5 and it led to 

efficient mixing than that the configurations in which tuyeres were placed at the PCD of 0.7 

and 0.59. The contributions of different mixing mechanism and the role of primary and 

secondary flow in determining the dynamics of gas-liquid flow and local mixing were studied. 

The modified bottom tuyere arrangements (referred as STAR-configurations) that create the 

flow variation in r-direction (primary flow) and also in θ-direction (secondary flow) were 

proposed. The flow rate variation along the radial and circumferential direction in an 

alternating arrangement generated re-circulatory cells in the circumferential direction (in 

addition to the re-circulatory cells in the radial direction) and this led to fast mixing inside the 

vessel. The mixing time was found to be reduced by 40-45 % than that of the conventional flow 

scheme. The experimentally verified computational model and the proposed STAR-type 

configuration will be useful to improve mixing efficiency, hence steel quality in BOF vessels. 
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सार 
 

 

 

वर्तमान थीसिि फैलाने वाले गैि-र्रल प्रवाह के मापने और िीएफडी सिमुलेशन के माध्यम िे 

बेसिक ऑक्सीजन फनेि (बीओएफ) के से्कल सकए गए ठंडे प्रवाह मॉडल में कई घूमने वाले बबल पू्लम्स 

द्वारा उत्पन्न र्रल चरण समश्रण की गसर्शीलर्ा की सवशेषर्ा पर कें सिर् है। प्राथसमक (r-सिशा में) और 

सद्वर्ीयक (θ- सिशा में) प्रवाह में घूमने वाले बबल पु्लम बीच जसिल इंिरैक्शन की भूसमका, जो बीओएफ 

वेिेल के अंिर स्थानीय समश्रण को सनयंसिर् करर्ी है, को प्रमासणर् सकया गया था। र्त्काल और िमय-

औिर् गैि वॉलू्यम अंश और स्थानीय समश्रण िमय को मापने के सलए इन-हाउि सवकसिर् प्रोब्स का 

उपयोग सकया गया है। यूलर-लग्रेंज (ईएल) अप्रोच का प्रयोग फैलाने वाले गैि-र्रल के ठंडे प्रवाह का 3 

डी क्षसणक सिमुलेशन के सलए सकया गया था। अनुमासनर् र्ात्कासलक और िमय-औिर् गैि वॉलू्यम अंश 

और स्थानीय के िाथ-िाथ स्थासनक रूप िे औिर् समश्रण िमय माप के िार् िंर्ोषजनक पाए गए थे। 

गैि-र्रल प्रवाह और समश्रण की गसर्शीलर्ा पर बबल पु्लम के अिंरु्सलर् व्यवहार का प्रभाव अध्ययन 

सकया गया था। सडफ़ाल्ट प्रवाह योजना (उद्योग में उपयोग की जाने वाली पारंपररक प्रवाह योजना) के 

मामले में, र्रल चरण प्रवाह के्षि अिमसमर् पाया गया था और इििे वेिेल के अंिर सवषम समश्रण होर्ा 

है। इि मामले में, स्थानीय समश्रण िमय एक स्थान िे िूिरे स्थान पर काफी सभन्न था। 

 गैि-र्रल प्रवाह और समश्रण की गसर्शीलर्ा पर मुक्त गैि-र्रल इंिरफेि के प्रभाव की जांच के 

सलए फैले हुए गैि-र्रल प्रवाह को वॉलू्यम-ऑफ-फु्लइड (वीओएफ) + यूलर-लैगे्रज (ईएल) सिमु्यलेशन 

और यूलर-लैगे्रज (ईएल) के िाथ सवसभन्न एच / डी अनुपार् में सकया गया । सनचले एच / डी अनुपार् के 

सलए, ईएल सिमुलेशन ने नो-स्लिप और फ्री-स्लिप िीमा पररस्लस्थसर्यो ं के िाथ प्रिशतन सकया, सजििे 

ऑिीलेिररी बबल पू्लम्स व्यवहार, उच्च र्रल वेग वाले के्षिो ंका कारण बन गया जो बिले में कम समश्रण 

िमय का कारण बना। इिके सवपरीर्, ईएल + वीओएफ सिमुलेशन ने बबल पू्लम्स और कम र्रल वेगो ं

की घूमने वाली गसर् को कम कर सिया सजिके पररणामस्वरूप समश्रण िमय बढ़ गया। सिलचस्प बार् यह 
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है सक ईएल + वीओएफ दृसिकोण का उपयोग करके अनुमासनर् समश्रण िमय माप के िाथ एक अचे्छ 

िमझौरे् में पाया गया था 

 पीिीडी (सपच-िू िकत ल-व्याि अनुपार्) का प्रभाव, जो रेसडयल सिशा में प्रवाह सभन्नर्ा प्रिान 

करर्ा है का  गैि-र्रल प्रवाह और समश्रण की गसर्शीलर्ा पर अध्ययन सकया गया था। यह पाया गया सक 

प्रवाह िमसमर् था जब आठ गॅि ईंलेि 0.5 के पीिीडी पर रखे गए थे और इििे कॉस्लफफ़गरेशन में उनीफ़ोमत 

समश्रण हुआ, 0.7 और 0.59 के पीिीडी की रु्लना में । गैि-र्रल प्रवाह और स्थानीय समश्रण की गसर्शीलर्ा 

को सनर्ातररर् करने में सवसभन्न समश्रण रं्ि और प्राथसमक और माध्यसमक प्रवाह की भूसमका का अध्ययन 

सकया गया। िंशोसर्र् िायर व्यवस्था (स्टार-कॉस्लफफ़गरेशन के रूप में िंिसभतर्) जो आर-सिशा (प्राथसमक 

प्रवाह) में प्रवाह सभन्नर्ा और θ-सिशा (माध्यसमक प्रवाह) में भी प्रवाह सभन्नर्ा का प्रस्ताव सिया गया था। 

इि  वैकस्लिक व्यवस्था में रेसडयल और पररर्ीय सिशा के िाथ प्रवाह िर सभन्नर्ा पररर्ीय सिशा में पुन: 

पररिंचरण उत्पन्न होर्ा है (रेसडयल सिशा में पुन: पररिंचरण के असर्ररक्त) और इििे वेिेल के अंिर 

रे्जी िे समश्रण होर्ा है। परंपरागर् प्रवाह योजना की रु्लना में समश्रण िमय 40-45% कम पाया गया। 

प्रयोगात्मक ित्यासपर् कम्प्यूिेशनल मॉडल और प्रस्तासवर् स्टार-प्रकार कॉस्लफफ़गरेशन समश्रण िक्षर्ा में 

िुर्ार के सलए उपयोगी होगा। 
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